# Components of Selection in the Evolution of the Influenza Virus: Linkage Effects Beat Inherent Selection

# Christopher J. R. Illingworth\*, Ville Mustonen\*

Wellcome Trust Sanger Institute, Hinxton, Cambridge, United Kingdom

## Abstract

The influenza virus is an important human pathogen, with a rapid rate of evolution in the human population. The rate of homologous recombination within genes of influenza is essentially zero. As such, where two alleles within the same gene are in linkage disequilibrium, interference between alleles will occur, whereby selection acting upon one allele has an influence upon the frequency of the other. We here measured the relative importance of selection and interference effects upon the evolution of influenza. We considered time-resolved allele frequency data from the global evolutionary history of the haemagglutinin gene of human influenza A/H3N2, conducting an in-depth analysis of sequences collected since 1996. Using a model that accounts for selection-caused interference between alleles in linkage disequilibrium, we estimated the inherent selective benefit of individual polymorphisms in the viral population. These inherent selection coefficients were in turn used to calculate the total selective effect of interference acting upon each polymorphism, considering the effect of the initial background upon which a mutation arose, and the subsequent effect of interference from other alleles that were under selection. Viewing events in retrospect, we estimated the influence of each of these components in determining whether a mutant allele eventually fixed or died in the global viral population. Our inherent selection coefficients, when combined across different regions of the protein, were consistent with previous measurements of dN/dS for the same system. Alleles going on to fix in the global population tended to be under more positive selection, to arise on more beneficial backgrounds, and to avoid strong negative interference from other alleles under selection. However, on average, the fate of a polymorphism was determined more by the combined influence of interference effects than by its inherent selection coefficient.

Citation: Illingworth CJR, Mustonen V (2012) Components of Selection in the Evolution of the Influenza Virus: Linkage Effects Beat Inherent Selection. PLoS Pathog 8(12): e1003091. doi:10.1371/journal.ppat.1003091

Editor: Raul Andino, University of California San Francisco, United States of America

Received July 10, 2012; Accepted November 5, 2012; Published December 27, 2012

**Copyright:** © 2012 Illingworth, Mustonen. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This work was supported by the Wellcome Trust under grant reference 098051. This research was supported in part by the National Science Foundation under Grant No. NSF PHY05-51164 during a visit at the Kavli Institute of Theoretical Physics (KITP, Santa Barbara, CA). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: ci3@sanger.ac.uk (CJRI); vm5@sanger.ac.uk (VM)

## Introduction

The influenza A virus is an important human pathogen, infecting between 5 and 15% of the global population each year [1]. Infection by influenza leads to strain-specific immunity in the human population, driving a rapid process of adaptation in the virus [2], particularly in the viral surface proteins haemagglutinin (HA) and neuraminidase (NA) [3].

Influenza evolves according to two distinct processes. Firstly, variation within gene sequences arises through a high mutation rate [4], the resulting mutations being subject to selection pressure. For HA and NA, the host immune system is a key driver behind selection [5–8], but selection can also arise from interactions with other viral genes [9], stability of the protein structure [10], adaptation to new species [11], drug resistance [12], and, in HA, glycosylation [13]. In a second process, known as reassortment, genes from viruses co-infecting a host combine to produce novel viral strains [14].

Many previous studies have measured the importance of selection in the evolution of influenza. Constructing a dendrogram from viral sequences, and evaluating occurrences of synonymous and non-synonymous substitutions, the ratio dN/dS has been used to identify positive selection, both at given loci, and in the HA

gene as a whole [15–17]. Other attempts to measure selection have looked for changes in frequencies, or in the distribution of frequencies, of nucleotides or amino acids, or rates of substitution between them, in individual loci, or gene-wide [18–22]. Each of these methods has produced estimates of selection based upon sets of multiple substitutionary events, either within single loci, or across a set of loci within a gene, rather than by considering single polymorphisms one by one.

Homologous recombination in influenza is extremely rare, if not entirely non-existent [23,24]. As such, within the viral population, interference effects, by which selection on one allele affects the frequency of another, are likely to occur [25–28]. On the basis that mutations observed at antigenic sites are more likely to be fixed than those at non-antigenic sites, it has been argued that the effects of hitchhiking in influenza are small [29]. However, evidence supporting the existence of clonal interference has been identified from the timings of fixation events in the HA gene, and from the inferred number of co-existent strongly beneficial mutations [30].

Here, we consider the evolution of the HA gene of the human influenza A/H3N2 strain since its emergence in 1968. H3N2 evolves more rapidly than H1N1 [3], having acquired multiple changes in amino acids over time. While reassortment between genes has occurred in this strain [31], the HA sequences represent

## **Author Summary**

Success in life is the product of many factors. Inherent ability often underlies great achievement. But other factors may play their part. The circumstances a child is born into may help or hinder his or her progress. Later events also have their effect; a life may be influenced by a lucky break, or an unforeseen disaster. In this work, we examine the factors underlying success for mutations in the HA gene of human influenza virus A/H3N2, defining success as the attainment of a high frequency in the global population. We examined the history of the gene from 1968 until 2010. For each observed mutation, a mathematical model was used to estimate the inherent benefit or disadvantage it conferred to the virus. We calculated the advantageousness or otherwise of the background upon which it arose, and the subsequent effect of interference from other mutations under selection. We found that successful mutations tended to have an advantageous background, and were subsequently fortunate in avoiding negative events throughout their lifetime. Beneficial mutations were more likely to be successful. But a mutation's chances of success were influenced more by circumstances of birth and subsequent events, than by its inherent effect on the virus.

a consistent strand of evolution, described by a large number of sequences [32]. We converted sequences of the HA gene into timeresolved measurements of allele frequencies at polymorphic loci, using these data to infer the strength of selection acting upon each allele.

Inference of selection coefficients was carried out using adapted forms of two methods, described in a previous publication [33], and illustrated in Figure 1. Collected frequencies of an allele, from its first observation until its death or fixation, are referred to as the allele trajectory. Firstly, an "unlinked" method, representing a minimal model for inferring selection, was applied. The selection acting upon an allele was measured in terms of a selection coefficient  $\sigma$ , equal to the difference in Malthusian fitness between it, and the wild-type allele at the same locus. For each trajectory *i*, a single selection coefficient,  $\sigma_i$ , was learnt, along with a single frequency,  $q_i^1$ , for some time-point within the trajectory (see Methods), so as to maximise the fit between the model, and the observed trajectory frequencies. Secondly, a "linked" model was applied, which used observed two-locus haplotype frequencies to account for interference effects between trajectories.

In the linked model, for each trajectory i, two selection coefficients were considered. The inherent selection coefficient,  $\sigma_i$ , again denotes the inherent advantage or disadvantage conveyed by the allele, relative to the wild-type allele, to individuals which possess it. The effective selection coefficient,  $\sigma_i^{\text{eff}}$ , by contrast, denotes the mean total advantage or disadvantage experienced by individuals with that allele, when selection effects acting upon alleles at all polymorphic loci are considered. This effective selection coefficient is time-dependent, and is written  $\sigma_i^{\text{eff}}(t_{i_k})$ , where  $t_{i_k}$  is the  $k^{\text{th}}$  time-point of the trajectory *i*. Given a set of selection coefficients for all trajectories,  $\{\sigma_i\}$ , our method calculates values of  $\sigma_i^{\text{eff}}(t_{i_k})$  for each trajectory. Under the model, allele frequencies at each locus change deterministically according to these effective selection coefficients. Selection coefficients,  $\sigma_i$ , and allele frequencies,  $q_i^1$ , were optimised to give the maximum likelihood fit between the model, and the observed allele frequencies.

Our "linked" model represents a minimal model of selection accounting for interference effects; while it is more complex than the unlinked model, it has no additional complexity, in the sense of additional parameters to be learnt. Rather, the additional component of the model derives from the underlying sequence data. Two-locus haplotype frequencies are measured at each timepoint. Next, if some allele a is under selection, and is in linkage disequilibrium with another allele b, selection acting on a changes the frequency of the allele b in accordance with that linkage disequilibrium. Due to its minimal nature, our linked model has some limitations. For example, the inherent selection coefficients,  $\sigma_i$ , do not change with time. Polymorphic alleles were assumed to interact according to a model of additive fitness. Further, to simplify the calculation, synonymous trajectories were assumed to have zero inherent selection, moving only via linkage disequilibrium with non-synonymous trajectories. Limitations of the model are discussed later in the text.

Application of our model gave an insight into the processes driving the evolution of influenza. Recent theoretical studies of the evolution of asexual populations have underlined the importance of the background upon which polymorphisms arise, and of interference from other polymorphisms under selection [34,35]. Experimental work has shown that, in a non-recombinant population, the background upon which a mutation lands can have a significant impact on its eventual fate [36]. On the basis of our inferences, we measured the importance of inherent selection, the initial genetic background, and subsequent interference effects, for the eventual fate of a polymorphism. Our results suggest that background effects, subsequent interference, and inherent selection are of similar importance for determining the fate of polymorphisms in the HA gene of human influenza A/H3N2. The combined effect of interference outweighs that of inherent selection.

#### Results

#### Summary statistics

A total of 3327 complete sequences of the HA gene of human influenza A/H3N2 were obtained from the NCBI influenza virus resource [32]. Trajectories reaching a frequency of at least 2.5% were included in the analysis (for discussion of this choice see Supporting Information), giving a total of 1638 trajectories (655 non-synonymous, 983 synonymous), spanning 256 time-points.

In the unlinked method, a selection coefficient and a single allele frequency were learnt for each trajectory, so as to obtain the maximum likelihood fit between the model, and the observed allele frequencies. In the linked method, synonymous trajectories were assumed to evolve under zero inherent selection, so for these, only an allele frequency was inferred. In the application of the linked method, pairwise interactions between all simultaneously polymorphic trajectories were accounted for; a mean of 4939 such pairs existed at any one given time-point.

One limitation in the inference arose from a lack of available sequences at early time points (see Supporting Information). As a result only polymorphisms which arose from 1996 onwards were included in the final analysis (spanning 84 time-points). The statistics that follow encompass 622 trajectories, of which 238 were non-synonymous, representing polymorphisms across 442 nucleotide positions (at 339 amino acid residues) in the HA gene. For these trajectories each observed allele frequency was calculated from a sample of mean size 202 sequences (range 57 to 449).

#### Linked and unlinked models of evolution

Including the effects of linkage disequilibrium between polymorphisms allowed the linked model to capture much of the





**Figure 1. Outline of methods for inferring selection. (A) Unlinked method** Above: Polymorphisms are considered in isolation. Two parameters are fitted to describe the evolution of the polymorphism over time, a selection coefficient  $\sigma_i$ , and a single allele frequency  $q_i^1$ . Below: Application of the method. Allele frequencies are measured over time (blue dots). The parameters are learnt independently of changes in other allele frequencies. Parameters which best reproduce the observed frequencies, under a likelihood model, are found (model frequencies shown in red). (**B**) **Linked method** Above: Polymorphisms are considered together, with two-locus haplotype frequencies being calculated from sequence data. Under an additive assumption, the effective selection on the A allele at locus *a* depends on selection at locus *b*, according to the equation shown (middle). Here only two polymorphic loci are illustrated; in reality selection at multiple polymorphic loci would influence the evolution of A. The effective selection  $\sigma_i^{\text{eff}}$  acting on A is time-dependent. It is re-calculated at each sampling point (green dots), and assumed to remain constant between sampling points (green line). The green dotted line shows the inherent selection  $\sigma_i$ . Below: The linked method is applied similarly to the unlinked method. However, all parameters are learnt simultaneously, selection on one allele potentially affecting the evolution of many others. Accounting for interference between alleles more faithfully reproduces the observed allele frequencies. Frequencies and parameters shown here are illustrative. doi:10.1371/journal.ppat.1003091.g001

complex behaviour of the observed trajectories. Relative to the unlinked approach, the model gave a substantially better fit to the observed trajectories (Figure 2 gives one example). In the unlinked model, the inference of selection coefficients for synonymous trajectories required, for the trajectories from 1996 onwards, an additional 384 parameters to be learnt in the optimisation. Nevertheless, the unlinked model gave a substantially worse fit to the data. Measured across more than 7000 observed frequencies,



Figure 2. Modelling linkage disequilibrium allows the fitting of complex trajectories. Inferences for a trajectory describing a polymorphism from ATT to AAT in locus 161. The trajectory inferred by the "linked" method (solid red line) gives a substantially better fit to the observed trajectory (blue dots), than does the trajectory inferred by the "unlinked" method (dashed green line). Including the effect of linkage disequilibrium between alleles gives time-dependent effective selection coefficients  $\sigma_i^{\text{eff}}(t_{i_k})$  that capture key features of the observed trajectory.

doi:10.1371/journal.ppat.1003091.g002

the mean absolute difference between inferred and observed allele frequencies was larger, at 0.09, for the unlinked model, than the equivalent value of 0.05 from the linked model. Where linkage disequilibrium is ignored, inferred frequencies are restricted to change according to a model of constant selection, giving a monotonically increasing or decreasing trajectory. Many trajectories encompassed both rising and falling allele frequencies, a phenomenon which, in a deterministic framework, requires a model of variable selection. Plots of all trajectories, together with model fits, are given in Supporting Information. We now discuss results obtained using the linked method.

#### Inherent selection and interference effects

Our model suggested interference effects to be of substantial importance in determining the eventual fixation or death of mutations in the influenza virus. For each trajectory *i*, a measure,  $\bar{\sigma}_i^{\text{eff}}$  of the total selection acting upon the mutant allele (including

both inherent effects and those arising from linkage disequilibrium) was calculated, as the mean value over time of the effective selection coefficient:

Mean effective selection : 
$$\bar{\sigma}_i^{\text{eff}} = (1/K_i) \sum_{k=1}^{K_i} \sigma_i^{\text{eff}}(t_{i_k}),$$
 (1)

where  $K_i$  is the length of trajectory *i*. Considering sets of polymorphic alleles at given time-points, substantial differences between  $\bar{\sigma}_i^{\text{eff}}$  and  $\sigma_i$  were seen (see for example Figure 3A). Both clonal interference and genetic hitchhiking were evident, with mutations under strong positive inherent selection, but negative interference, dying out, and mutations with lesser inherent selection, but experiencing positive interference, reaching fixation. Both in the examples shown, and across all trajectories (Figure 3B), the statistic  $\bar{\sigma}_i^{\text{eff}}$ , inferred after the event, discriminated strongly between mutations that fixed and those which died out.

## Components of selection

Derivative components of selection, associated with interference effects, were calculated for each of the trajectories. Firstly, we derived a measure of the mean effect, over time, of interference acting upon each trajectory. We note that the mean effective selection coefficient,  $\bar{\sigma}_i^{\text{eff}}$ , calculated above, combines the inherent selection acting on a trajectory,  $\sigma_i$ , plus interference effects from other trajectories. We calculated a measure,  $\bar{\sigma}_i^{\text{t}}$ , of the mean interference acting on a trajectory *i* by taking the difference of the mean effective and inherent selective effects:

Mean total interference : 
$$\bar{\sigma}_i^t = \bar{\sigma}_i^{eff} - \sigma_i$$
. (2)

This interference term can be divided into two parts, representing the effects of interference in the initial and later parts of the history of a polymorphism. Considering these in turn, any given mutation observed in the population is first seen in the context of a specific sequence background. Within the global viral population at any one time, non-synonymous polymorphisms at frequencies considered by our method (reaching at least 2.5%) exist at multiple loci. The sequence(s) upon which a mutation arises contain specific alleles at these loci, this background conferring an initial fitness effect. We estimated this background fitness for a trajectory i,



**Figure 3. Inferred and mean effective selection coefficients show substantial interference effects.** (**A**) a snap-shot:Values of  $\sigma_i$  (circles) and  $\bar{\sigma}_i^{\text{eff}}$  (squares) inferred for trajectories which were polymorphic in the sample taken on 29th October 2002, ordered by  $\sigma_i$ . Substantial differences are seen between the inferred and mean effective selection coefficients, demonstrating significant interference effects. Interference is seen to have a substantial effect on the fate of polymorphisms; all trajectories which proceed to fixation (red) have  $\bar{\sigma}_i^{\text{eff}}$  above the dashed horizontal line, while all trajectories which eventually die out (blue) have  $\bar{\sigma}_i^{\text{eff}}$  below the line. (**B**) all data since 1996: Histogram of  $\bar{\sigma}_i^{\text{eff}}$  values for trajectories, those which died being shown in blue and those that proceeded to fixation being shown in red. doi:10.1371/journal.ppat.1003091.g003

denoted  $\sigma_i^b$ , as the mean interference over the period before the trajectory reached a frequency of 2.5%:

Initial background fitness : 
$$\sigma_i^b = [(1/r) \sum_{k=1}^r \sigma_i^{\text{eff}}(t_{i_k})] - \sigma_i,$$
 (3)

where  $t_{i_r}$  denotes the first time at which the allele of trajectory *i* was observed at a frequency of 2.5% or greater. Subsequently, across the lifetime of a trajectory, its evolution is affected by the arrival of new mutations under selection. Changes also occur in the effect of its background; as the mean fitness of the population increases, any given background becomes relatively less beneficial. We measured the combined effect of these factors, terming it the post-emergence interference,  $\bar{\sigma}_i^p$ . This statistic was calculated by subtracting the initial background effect from the total interference:

Mean postemergence interference : 
$$\bar{\sigma}_i^p = \bar{\sigma}_i^t - \sigma_i^b$$
 (4)

We note that each of the above components of selection are derived from the inferred selection coefficients  $\sigma_i$ , rather than being learnt independently from the data.

Across trajectories, the magnitude of the effect of interference was generally greater than that of inherent selection. Measured for non-synonymous trajectories, the mean absolute effect of interference acting on a polymorphism at a given time was larger, at 9.7, than the mean absolute inherent selection of 8.0 (selection is described throughout in units of  $10^{-3} \text{ day}^{-1}$ ). Simple characterisation of interference in terms of a few large effects was difficult; at any given time, a mean of 15 other trajectories made an absolute contribution of more than 0.5 to the effective selection coefficient

of a polymorphism, from a total of 29 trajectories contributing more than 0.1.

To examine selective effects in more detail, we calculated distributions of each statistic for trajectories that eventually fixed, and for those that died out (Figure 4 shows data for nonsynonymous trajectories). In each case, a bias can be seen, trajectories proceeding to fixation having generally more positive coefficients of selection and interference. Different components of selection biased the fates of trajectories in different ways. For example, the initial background for observed trajectories was generally positive, being more so for fixing trajectories, while the mean post-emergence interference was predominantly negative. Fixing mutations tended to arise in good backgrounds, and were subsequently fortunate in avoiding strong negative interference.

In order to quantify the importance of each of the components of selection for the fate of polymorphisms, we measured the accuracy with which each component, considered in retrospect, separated trajectories which reached fixation from those that died out. Across non-synonymous trajectories, the inherent selection coefficient showed some ability to perform this task, with a calculated accuracy of 0.70 (Figure 5). Both the initial background fitness, and the post-emergence interference, performed similarly well, with measured accuracies of 0.64 and 0.70. When interference effects were combined, the accuracy achieved was greater than that for inherent selection, returning a value of 0.82.

Across synonymous trajectories, the inherent selection coefficient was set by default to zero, such that the effect of any inherent selection could not be measured. However, for these trajectories, the accuracies of the initial background and post-emergence interference were again roughly equivalent, at 0.75 and 0.71 respectively. The evolution of synonymous trajectories was wellexplained purely in terms of interference from non-synonymous polymorphisms under selection; absolute differences between



**Figure 4. Components of selection: breakdown of selection and interference effects.** Histograms show the total distribution of values obtained for non-synonymous trajectories, trajectories which died shown in blue and which proceeded to fixation in red. (**A**) Inherent selection coefficients  $\sigma_i$ ; (**B**) Initial background fitness  $\sigma_i^b$ ; (**C**) Mean post-emergence interference  $\bar{\sigma}_i^p$ ; (**D**) Mean total interference  $\bar{\sigma}_i^t$ . Distributions are calculated across a total of 218 trajectories. Fixing trajectories tended to appear on more positive backgrounds, and subsequently avoided strong negative interference.

doi:10.1371/journal.ppat.1003091.g004



**Figure 5.** Ability of retrospective measurements of selection to divide fixing from non-fixing trajectories. Values are shown for nonsynonymous and synonymous polymorphisms, representing statistics for the inherent selection coefficient,  $\sigma_i$  (blue), the initial background fitness,  $\sigma_i^{\rm b}$  (green), the mean post-emergence interference,  $\sigma_i^{\rm p}$  (yellow), the total mean interference,  $\sigma_i^{\rm t}$  (orange), and the mean effective selection coefficient,  $\sigma_i^{\rm eff}$  (red) of a trajectory. The black dotted line indicates a null expectation, describing a hypothetical method with no power to divide fixing from non-fixing trajectories. For synonymous polymorphisms, inherent selection coefficients were set to zero. As such, the measures for inherent and mean effective selection coefficients are omitted, the former being equal by default to the null expectation, and the latter being equal to the total mean interference.

doi:10.1371/journal.ppat.1003091.g005

observed and inferred frequencies were no larger than those for non-synonymous trajectories.

## Selection across the HA gene

Measuring the relative strengths of selection acting upon different parts of the HA protein, we obtained results equivalent to those previously obtained from measurements of dN/dS, the ratio between rates of non-synonymous and synonymous mutations. We derived a region-wide measure of selection by calculating the mean of the inherent selection coefficients inferred for non-synonymous trajectories at loci in that region. This measure was then calculated for trajectories in the HA1 and HA2 segments of the protein, and for both epitope and non-epitope loci. Substitutions in the influenza phylogenetic tree can be divided into those occurring on trunk and side branches [37]; to parallel this, we calculated separate measurements for trajectories which fixed in the global population (c.f. trunk substitutions), and for those which died out (c.f. side-branch substitutions). A previous calculation performed on substitutions in a tree of sequences from the human influenza A/H3N2 strain between 1994 and 2005 found dN/dS ratios fitting the order epitope >HA1>HA2, with ratios in each case higher for substitutions in trunk branches than in non-trunk branches [17]. Our measurement replicated this finding across corresponding sets of trajectories (Table 1).

 Table 1. Mean inherent selection coefficients inferred by the method.

Protein sites	Mean inherent selection			
	All trajectories	Fixing trajectories	Dying trajectories	
HA	1.12	6.80	0.04	
HA1	1.33	7.71	0.10	
HA2	0.18	2.38	-0.22	
Epitope	2.29	8.53	0.54	
Non-epitope	-0.15	2.47	-0.43	

Mean values of  $\sigma_i$  inferred for non-synonymous trajectories within the HA1 and HA2 parts of the gene, and for epitope and non-epitope residues, considering both fixing and dying trajectories.

doi:10.1371/journal.ppat.1003091.t001

The levels of significance underlying our results were calculated by applying a Kolmogorov-Smirnov test to to the inferred distributions of selection coefficients; differences in inferred selection coefficients between epitope and non-epitope trajectories, and between the subclasses of fixed epitope and fixed non-epitope trajectories were each significant (p = 0.014 and p = 0.021respectively). A significant difference was seen between selection coefficients at epitope and HA2 trajectories (p = 0.030), though differences between trajectories in HA1 and HA2 were not significant at the 95% level. In our calculation, epitope loci were identified according to the alignment of Wolf et al. [17].

The difference between epitope and non-epitope loci was also seen via measurements of fitness flux, a statistic quantifying the total amount of adaptation at a locus [38]. Across trajectories reaching fixation or death, the total fitness flux for a trajectory *i* is given simply by  $\Phi_i = \sigma_i$  if the trajectory fixed, and  $\Phi_i = 0$  if the trajectory died. On the basis of our inferred selection coefficients, the total fitness flux across trajectories at epitope loci was approximately 8.6 times larger than the flux across trajectories at non-epitope loci. While both beneficial and deleterious mutations were observed in all parts of the gene, this result implies that events at epitope loci are of substantial importance for the adaptation of the virus.

Regions in which beneficial mutations were identified correlated with those under lesser evolutionary constraint. The extent of evolutionary constraint was measured by examining the number of non-synonymous polymorphisms observed across different sets of multiple loci. Supposing a constant mutation rate across the gene, and lack of any constraint, two regions each comprising a similar proportion of the gene would be expected to harbour a similar proportion of non-synonymous mutations. In regions with an increased constraint on mutations, as might arise from requirements of protein function or structural stability, a greater number of these mutations would be strongly deleterious, leading to a relative under-representation of non-synonymous trajectories from that region. In our data, the HA2 region, and, to a lesser extent, the non-epitope region of HA1, were each underrepresented for observed polymorphisms and trajectories, while the HA1 region as a whole, and in particular, epitope loci, were overrepresented. This pattern was exaggerated for trajectories identified to be under positive selection ( $\sigma_i > 0$ ), and further so for trajectories with  $\sigma_i > 5$ (Table 2), suggesting that sites experiencing greater numbers of beneficial mutations, were also sites of lesser evolutionary

Table 2. Over- and under-representation of trajectories in different regions of HA.

Protein sites	Number of events relative to size					
	All polymorphisms	All trajectories	Trajectories ( $\sigma_i > 0$ )	<b>Trajectories</b> ( $\sigma_i > 5$ )		
HA1	1.29***	1.40***	1.46***	1.50***		
HA2	0.59***	0.43***	0.36***	0.30***		
Epitope	2.05***	2.40***	2.60***	2.97***		
HA1 non-epitope	0.84***	0.81*	0.77*	0.63**		

Proportion of trajectories occurring in different regions of HA. Values are expressed relative to the size of each region, measured by number of residues. The set of all polymorphisms includes trajectories excluded from the analysis (i.e. with maximum frequency less than 2.5%). Deviations from the expected value of 1 are labelled according to their significance:

doi:10.1371/journal.ppat.1003091.t002

constraint. This result corresponds to the HA protein structure, in which epitope residues tend to be in regions more exposed regions of the protein [39].

## Discussion

The problem of inferring the selection acting at every polymorphism within the history of the human influenza A/ H3N2 HA gene represents a significant challenge. Here we have examined the role of interference, as opposed to inherent selection, in determining the eventual fixation or death of polymorphisms. We identified, first, that interference is of great importance in understanding the evolution of influenza. While having no more parameters than the unlinked model, our linked inference model produced trajectories that substantially better fitted observed allele frequencies over time. Next, examining statistics of selection from our linked model, we inferred that, for non-synonymous polymorphisms, the total effect of interference acting on a polymorphism was, in the mean, more significant for its fate than was its inherent selection coefficient. With the inherent selection acting on synonymous polymorphisms expected to be substantially weaker than that acting on non-synonymous polymorphisms (see discussion later), our result is likely to hold for polymorphisms in the HA gene in as a whole.

For both non-synonymous and synonymous events, the effect of interference was split roughly equally between background effects, and subsequent interference. While we do not make specific estimates of population size, mutation rate, or the distribution of fitness effects (we infer selection coefficients only for polymorphisms which are observed at above a given frequency), the key role we observe for background and interference effects is in line with mathematical expectations for an asexual system with high rates of mutation and positive selection [34,35].

While we replicate results previously obtained using a measure of dN/dS, our approach to inferring selection is substantially different. For example, whereas dN/dS comprises a statistic, calculated across multiple events, we attempt to infer a selection coefficient for each individual polymorphism. Further differences lie in our use of time-resolved frequency data; rather than inferring a phylogeny we consider sequences in terms of their allele and haplotype frequencies. A trajectory-based approach may differ from a phylogeny in terms of the number of times a mutation is observed; multiple identical substitutions in different parts of a tree might comprise one or multiple trajectories. However, several features in the H3N2 HA gene that have been identified to be under selection, including

repeated mutations at a site, selective sweeps, and highly polymorphic sites [20], were visible in our set of allele trajectories. We note that, where sequences are taken from a single population, the interpretation of values of dN/dS is not always straightforward [40].

Our inference of strong interference effects has implications for the evolution of influenza. Firstly, interference between beneficial mutations slows down the overall rate of adaptation [25]. The lack of recombination within genes of influenza means that beneficial mutations compete against each other for success. A mutation causing the virus to escape the immunity of the host population might be outcompeted by another. The most inherently advantageous mutations are not guaranteed to fix. Further, interference makes the prediction of viral evolution more difficult. Identifying the most beneficial mutations at a given time does not necessarily identify those that will go on to fixation. A polymorphism that was growing in frequency might be killed off by the emergence of another, more beneficial mutation. A polymorphism that existed at low frequency, with no apparent benefit for the virus, might, through hitchhiking, reach fixation.

#### Model limitations

As noted above, our linked model represents a minimal approach for inferring selection while accounting for interference effects. While our model provides strong evidence of the importance of interference effects, it does not account for the full biological complexity of influenza, making a number of assumptions and approximations.

Firstly, synonymous mutations were assumed to have zero inherent selection coefficient, synonymous trajectories changing their frequency purely by the effect of linkage disequilibrium with non-synonymous trajectories. Such an assumption is not unique to our method, reflecting similar assumptions made, for instance, in the measurement of dN/dS. However, the identification in HA of some selection for RNA secondary structure [41] and of long-term changes in codon preference [19] each suggest the possibility that some synonymous changes are non-neutral. This shortcoming in our method is unlikely to affect our central result, of the importance of interference effects. While individual synonymous mutations may be under significant selection, the magnitude of selection, considered across groups of trajectories, is likely to be substantially lower for synonymous than non-synonymous mutations. As such, any errors in our statistics of selection arising from this assumption are likely to be small when considered across the set of all trajectories.

<sup>\*</sup>p<0.05;

<sup>\*\*</sup>p<0.01; \*\*\*p<10<sup>-6</sup>

Secondly, in our use of a deterministic model for the evolution of trajectories, we do not take into account genetic drift, a process, independent of selection, in which the transmission of alleles from one generation to the next introduces noise into the allele frequencies. (Within our framework, the frequency of an allele evolving under zero selection would remain constant.) We believe that the effect of this omission is also likely to be small. Considering alleles at intermediate frequencies, we note that genetic drift is a slow process. In influenza it has been estimated that the expected time for an allele to fix under drift alone is of the order of 200 years, far longer than the typical length of a trajectory [29], implying that frequency changes are driven much more by selection than by drift. As evidenced by antigenic drift, strongly selected alleles arise frequently [6], and, due to the lack of recombination, are likely to remain in linkage disequilibrium with other alleles at polymorphic loci. Interference arising from these strongly selected alleles would mean that changes in the frequencies of neutral alleles would also be dominated by selection [34]. Considering alleles at lower frequencies, the influence of selection on an allele is overtaken by that of drift below a frequency of  $1/N\sigma$ , where N is the population size and  $\sigma$  the magnitude of selection acting on the allele [42]. In recent work, an estimate has been made that in H3N2 influenza this threshold is close to 1% [30]. Across our observed trajectories, one out of 12 observed frequencies were below this threshold. If this threshold estimate is correct, our exclusion of trajectories with minimal frequencies less than 2.5% ensures that the majority of our trajectories are influenced more by selection than drift for the majority of their lifetimes.

Thirdly, the inferences we make are drawn on the basis of limited data. While we filter trajectories to consider only those for which we had enough data to derive consistent results, studies of simulated systems suggest that a larger sample would lead to more accurate inference. Consideration of lower frequency events may affect our estimate of the relative importance of the initial background and subsequent interference on a polymorphism; measuring at lower frequencies, more deleterious mutations, and mutations on more deleterious backgrounds, would likely be seen.

We further note that, while detecting selection, our method does not infer reasons for this selection. The molecular basis for the evolution of influenza is complex, involving multiple host-virus interaction processes, and the biophysical properties of the viral proteins. Each of these processes may result in changes in allele frequencies, and their mean effect can potentially be recaptured via inference. However, assignment of selection to specific physical effects would require substantial further work. Our approach is not well-suited to identifying the magnitude of selection arising from evolutionary constraints, such as protein structural stability; where an allele is under strong purifying selection, by its nature it is not likely to be observed at frequencies high enough for selection to be inferred.

Extensions to the linked model could be made to include the possibility of time-dependent inherent selection acting on a given allele, or of non-additive models of interference between trajectories. Either of these would add to the complexity of the model, and would require care in implementation.

Time-dependent inherent selection in influenza might arise through the acquisition of strain-specific immunity among the human population to an antigenic change in the virus [43,44]. A mutation causing a novel antigenic type might initially be at a selective advantage within the viral population. However, as people encountered the new antigenic type, and acquired immunity, this selective advantage would decrease. Variable selection corresponding to antigenic type has been included in models of influenza evolution by mapping changes in genetic sequence to changes in antigenic type [2,45,46]. Time-dependent selection could be incorporated into the model presented here either with the incorporation of antigenic data, or simply by allowing inherent selection coefficients to vary over time. However, we note, first of all, that time-dependent selection would not remove the role of interference. Linkage disequilibrium between alleles implies that the evolution of one allele is affected by any non-neutral selection acting on the other, whether or not that selection is time-dependent. Secondly, any adding of parameters to the model requires care to avoid over-fitting; in the limit case, allowing the inherent selection on each allele to vary between each time point would result in a perfect fit to the observations by default. The inherent selection coefficients inferred here are best interpreted as mean values over time.

Epistasis (non-additive selective effects) between alleles has been detected in the HA and NA genes of influenza by considering patterns of ordered substitution events [47]. Full inclusion of epistasis into a model, however, is difficult, due to the rapid inflation of the number of parameters required. With the addition of a substantial number of parameters, epistatic effects between pairs of mutations could potentially be modelled. However, there is no guarantee that epistasis operates purely on this level; interactions between three, four, or more mutations may also play a role in viral fitness. Our model includes epistatic effects where all but one of the mutations in question are fixed; repeated instances of identical mutations are assigned independent selection coefficients. We note that, under certain circumstances, epistasis may be irrelevant. Where two alleles are seen only in isolation in the population, and never together in combination, epistatic effects between them do not come into play. Beyond this, incorporation of epistasis into the model would be challenging, and was not attempted here, the assumption of additive fitness effects representing a first approximation.

#### Interference and viral evolution

Using a minimal model accounting for interference, we have here characterised selection in the HA gene of influenza at the level of individual polymorphisms. Our approach takes into account the fact that the gene evolves as part of a haploid, asexual virus with high mutation rate, and essentially no homologous recombination, so that linkage disequilibrium between polymorphisms has a substantial impact on its evolution. Examining statistics of selection inferred for historic data, our model suggests that the combined effect of interference is more important than the effect of inherent selection for the fixation or death of a polymorphism. While many methods for inferring selection exist, we believe that it is by accounting for the specific genetic properties of systems such as influenza that progress in understanding their evolution will most rapidly be made.

## Methods

#### Obtaining polymorphism data

Complete sequences for the HA gene of the subtype H3N2 of human influenza A were obtained from the NCBI influenza virus resource [32], including all sequences sampled from 1968 until 7th Feb 2011, a total of 3327 sequences. Sequence alignment of protein sequences was carried out using MUSCLE [48], corresponding RNA sequences being matched to this alignment using in-house software. Beginning at 29th June 1968, samples of sequences were taken every 60 days, each sample comprising all sequences recorded within 180 days of the sample point. Sampling in this manner grouped viral sequences taken from all seasons in both hemispheres, accounting for geographical and seasonal differences. Ambiguously dated sequences were assigned to the central day of their specified year or month of sampling.

## Calling trajectories

The frequency of a given allele was measured as the fraction of sequences within a sample having that allele at the time of measurement. Trajectories were defined as sets of observations of codon frequencies differing from the wild-type codon at their respective position. The wild-type codon for each position was initially defined according to the consensus codon from the first sequence sample; subsequent to this, a codon which reached fixation was defined as the new wild-type for that position. The identification of the point at which a trajectory has fixed or died is non-trivial. Within a single patient, the high mutation rate and large viral population size of influenza make it likely that every viable single-mutant of an infecting virus will likely exist, such that concepts of fixation and death require a qualified understanding. We regarded a trajectory as having fixed if, in our sample of the global population, we observed it at a frequency of greater than 97.5% for eight consecutive time points, and to have died if we observed a frequency of less than 2.5% for eight consecutive time points. The wild-type codon at a position was used to classify new mutations as synonymous or non-synonymous. Synonymous polymorphisms were set to have inherent selection coefficient equal to zero, changes in their frequency being inferred to arise through linkage disequilibrium with non-synonymous polymorphisms. In order to reduce the influence of noise from sampling error, and to avoid difficulties in inferring selection from very sparse data, trajectories with very low maximal frequencies (<2.5%) were excluded from the optimisation (see Supporting Information).

## Inferring inherent selection coefficients

In an earlier publication we described two methods for using time-resolved data for inferring the selection acting upon a polymorphic allele [33]. For notation, we describe a set of observations of the frequency of a polymorphic allele collectively as an allele trajectory. We consider a trajectory *i*, describing the evolution of a mutant allele at a two-allele locus *i'*, at which wild-type and mutant loci are denoted 0 and 1 respectively (separate annotation for trajectories and loci, while superfluous here, becomes necessary in the multi-allele case). The observed frequency of the mutant allele at locus *i'* at time *t* is denoted  $\hat{q}_{i'}^{i}(t)$ . We assume that the mutant allele has a constant, trajectory-specific selection coefficient,  $\sigma_i = f_{i'}^1 - f_{i'}^0$ , defined as the difference in Malthusian fitnesses of the two alleles at the given locus.

Our first model, referred to above as the "unlinked" model, assumes that alleles at different loci evolve independently of each other. Using the above notation, the expected frequency of the mutant allele at locus i' evolves according to the equation

$$\frac{dq_{i}^{1}}{dt} = \sigma_{i}q_{i}^{1}(1-q_{i}^{1}).$$
(5)

This equation describes a family of trajectories, specified by the allele frequency at time  $t_0$ :

$$q_{\vec{i}}^{1}(t) = \frac{q_{\vec{i}}^{1}(t_{0})e^{\sigma_{\vec{i}}(t-t_{0})}}{1-q_{\vec{i}}^{1}(t_{0})+q_{\vec{i}}^{1}(t_{0})e^{\sigma_{\vec{i}}(t-t_{0})}}.$$
(6)

In practise, the allele frequency at any time point may be used to specify curves; the central point of the trajectory, denoted  $t_{i_c}$ , was

used in our calculations. The model which best fits the observed trajectory was identified by optimising the selection coefficient and allele frequency (here  $\sigma_i$  and  $q_i^1(t_{i_c})$ ) using a binomial model to fit inferred frequencies  $q_{i'}^1(t)$  to the observed frequencies  $\hat{q}_i^1(t)$ .

Our second model, referred to above as the "linked" model, is essentially the same as this, but for the constraint that, if two alleles are in linkage disequilibrium, a change in the frequency of one allele, caused by selection, implies a change in the frequency of the other. At each time-point, two-locus haplotype frequencies, denoted  $\hat{q}_{ij}^{ab}(t)$  for loci i',j' and  $a,b \in \{0,1\}$ , were observed and used to calculate the effect of selection at one locus on the frequency of another.

Given a trajectory *i* representing the evolution of the allele 1 at locus *i'*, a time-dependent selection coefficient  $\sigma_i^{\text{eff}}(t)$  was calculated. This term describes the selection acting on the mutant allele of the trajectory given its own, or inherent selection coefficient,  $\sigma_i$ , and the selection acting as a result of linkage disequilibrium with alleles at other loci, which were also under selection, at the time *t*. Under a model of additive selection, this can be approximated by

$$\sigma_i^{\text{eff}}(t) = \sigma_i + \sum_{j \neq i} \sigma_{ij}(t), \tag{7}$$

where

$$\sigma_{ij}(t) = \sigma_j \left( \frac{\hat{q}_{ij}^{11}(t)}{\hat{q}_{ij'}^{11}(t) + \hat{q}_{ij'}^{10}(t)} - \frac{\hat{q}_{ij'}^{01}(t)}{\hat{q}_{ij'}^{01}(t) + \hat{q}_{ij'}^{00}(t)} \right).$$
(8)

The values  $\sigma_i^{\text{eff}}(t)$  were assumed to remain constant between sampling points. Under the model, these time-dependent selection coefficents were fed into Eq. (6), to describe the evolution of the trajectory; again a fit was performed between the inferred and observed frequencies. For simplicity, we have here described the case in which there are two alleles per locus. In the adaptation of our previously published method to influenza data, changes were made to account for the process via which trajectories were sampled, and to allow independent trajectories at the same locus to have differing selection coefficients. Further changes were made to the manner in which the fixation or death of trajectories was handled, to the treatment of very low frequency trajectories and of nonpolymorphic observations in the middle and ends of trajectories, to the precise use of effective selection coefficients in inferring trajectories, and to the fitting of inferred to observed trajectory frequencies. Fuller details of the method are given in Supporting information.

#### Supporting Information

**Text S1** Expanded description of the basic inference method. Further details of adaptations to the inference method conducted for the purposes of this study. Details of assessment of the method using simulated data. Description of the method for calling trajectories, and of an alternative method. Details of the optimisation routine. Details of the consistency of results obtained from the influenza dataset. Supporting Figures S1 to S13 are contained within the Supporting Text. (PDF)

**Figure S14** Plots of all observed and inferred trajectories inferred using the unlinked method. (PDF)

**Figure S15** Plots of all observed and inferred trajectories inferred using the linked method. (PDF)

**Figure S16** Plots of all observed and inferred trajectories inferred using the linked method with trajectories called using an alternative method. (PDF)

#### Acknowledgments

We would like to thank Andrej Fischer, Chris Greenman, and three anonymous referees for helpful comments on an earlier version of this

## References

- 1. World Health Organisation (2003). Fact sheet number 211. Inuenza.
- Ferguson N, Galvani A, Bush R (2003) Ecological and immunological determinants of inuenza evolution. Nature 422: 428–433.
- Bhatt S, Holmes EC, Pybus OG (2011) The genomic rate of molecular adaptation of the human inuenza A virus. Molecular biology and evolution 28: 2443–51.
- Parvin JD, Moscona A, Pan WT, Leider JM, Palese P (1986) Measurement of the mutation rates of animal viruses: inuenza A virus and poliovirus type 1. Journal of Virology 59: 377–83.
- Webster RG, Bean WJ, Gorman OT, Chambers TM, Kawaoka Y (1992) Evolution and ecology of inuenza A viruses. Microbiol Rev 56: 152–79.
- Smith DJ, Lapedes AS, de Jong JC, Bestebroer TM, Rimmelzwaan GF, et al. (2004) Mapping the antigenic and genetic evolution of inuenza virus. Science 305: 371–6.
- Rambaut A, Pybus O, Nelson M, Viboud C, Taubenberger J, et al. (2008) The genomic and epidemiological dynamics of human inuenza A virus. Nature 453: 615–619.
- Park AW, Daly JM, Lewis NS, Smith DJ, Wood JLN, et al. (2009) Quantifying the impact of immune escape on transmission dynamics of inuenza. Science 326: 726–8.
- Holmes EC, Ghedin E, Miller N, Taylor J, Bao Y, et al. (2005) Whole-genome analysis of human inuenza a virus reveals multiple persistent lineages and reassortment among recent H3N2 viruses. PLoS Biol 3: e300.
- Bloom JD, Gong LJ, Baltimore D (2010) Permissive secondary mutations enable the evolution of inuenza oseltamivir resistance. Science 328: 1272–5.
- 11. dos Reis M, Tamuri AU, Hay AJ, Goldstein RA (2011) Charting the host adaptation of inuenza viruses. Molecular biology and evolution 28: 1755–67.
- Scheell JR, Chou JJ (2008) Structure and mechanism of the M2 proton channel of inuenza A virus. Nature 451: 591–5.
- Das SR, Puigbò P, Hensley SE, Hurt DE, Bennink JR, et al. (2010) Glycosylation focuses sequence variation in the inuenza A virus H1 hemagglutinin globular domain. PLoS Pathogens 6: e1001211.
- Octaviani C, Ozawa M, Yamada S, Goto H, Kawaoka Y (2010) High level of genetic compatibility between swine-origin H1N1 and highly pathogenic avian H5N1 inuenza viruses. Journal of virology 84: 10918.
- Bush RM, Fitch WM, Bender CA, Cox NJ (1999) Positive selection on the H3 hemagglutinin gene of human inuenza virus A. Molecular biology and evolution 16: 1457–65.
- Yang Z, Nielsen R, Goldman N, Pedersen AM (2000) Codon-substitution models for heterogeneous selection pressure at amino acid sites. Genetics 155: 431–49.
- Wolf YI, Viboud C, Holmes EC, Koonin EV, Lipman DJ (2006) Long intervals of stasis punctuated by bursts of positive selection in the seasonal evolution of inuenza A virus. Biol Direct 1: 34.
- Kryazhimskiy S, Bazykin GA, Plotkin JB, Plotkin J, Dushoff J (2008) Directionality in the evolution of inuenza A haemagglutinin. Proc Biol Sci 275: 2455–64.
- Kryazhimskiy S, Bazykin GA, Dushoff J (2008) Natural selection for nucleotide usage at synonymous and nonsynonymous sites in inuenza A virus genes. Journal of virology 82: 4938–45.
- Pond SLK, Poon AF, Brown AJL, Frost SD (2008) A maximum likelihood method for detecting directional evolution in protein sequences and its application to inuenza A virus. Molecular biology and evolution 25: 1809–1824.
- Tamuri AU, Reis MD, Hay AJ, Goldstein RA (2009) Identifying changes in selective constraints: Host shifts in inuenza. PLoS Comput Biol 5: e1000564.
- Steinbrück L, Mchardy AC (2011) Allele dynamics plots for the study of evolutionary dynamics in viral populations. Nucleic Acids Research 39: e4.
- Boni MF, Zhou Y, Taubenberger JK, Holmes EC (2008) Homologous recombination is very rare or absent in human inuenza A virus. Journal of virology 82: 4807–11.
- Boni MF, Jong MDD, van Doorn HR, Holmes EC (2010) Guidelines for identifying homologous recombination events in inuenza A virus. PLoS ONE 5: e10434.

manuscript, and Michael Lässig for discussions and sharing a prepublication manuscript. We would like to thank participants of the KITP program on Microbial and Viral Evolution for discussions.

#### **Author Contributions**

Conceived and designed the experiments: CJRI VM. Performed the experiments: CJRI. Analyzed the data: CJRI. Contributed reagents/ materials/analysis tools: CJRI. Wrote the paper: CJRI VM. Interpreted the results: CJRI VM.

- Hill W, Robertson A (1966) The effect of linkage on limits to artificial selection. Genetical Research 8: 269–294.
- Smith J, Haigh J (1974) The hitch-hiking effect of a favourable gene. Genetical Research 23: 23–35.
- Miralles R, Gerrish PJ, Moya A, Elena SF (1999) Clonal interference and the evolution of RNA viruses. Science 285: 1745–7.
- de Visser JAAM, Zeyl CW, Gerrish PJ, Blanchard JL, Lenski RE (1999) Diminishing returns from mutation supply rate in asexual populations. Science 283: 404–6.
- Shih ACC, Hsiao TC, Ho MS, Li WH (2007) Simultaneous amino acid substitutions at antigenic sites drive inuenza A hemagglutinin evolution. Proc Natl Acad Sci U S A 104: 6283–8.
- Strelkowa N, Lässig M (2012) Clonal interference in the evolution of inuenza. Genetics 192: 671–82.
- Nelson MI, Simonsen L, Viboud C, Miller MA, Taylor J, et al. (2006) Stochastic processes are key determinants of short-term evolution in inuenza A virus. PLoS Pathogens 2: e125.
- Bao Y, Bolotov P, Dernovoy D, Kiryutin B (2008) The inuenza virus resource at the National Center for Biotechnology Information. Journal of virology 82: 596– 601.
- Illingworth CJR, Mustonen V (2011) Distinguishing driver and passenger mutations in an evolutionary history categorized by interference. Genetics 189: 989–1000.
- Schiffels S, Szöllosi GJ, Mustonen V, Lässig M (2011) Emergent neutrality in adaptive asexual evolution. Genetics 189: 1361–75.
- Good BH, Rouzine IM, Balick DJ, Hallatschek O, Desai MM (2012) Distribution of fixed beneficial mutations and the rate of adaptation in asexual populations. Proc Natl Acad Sci U S A 109: 4950–5.
- Lang GI, Botstein D, Desai MM (2011) Genetic variation and the fate of beneficial mutations in asexual populations. Genetics 188: 647–61.
- Fitch WM, Leiter JM, Li XQ, Palese P (1991) Positive Darwinian evolution in human inuenza A viruses. Proceedings of the National Academy of Sciences of the United States of America 88: 4270–4.
- Mustonen V, Lässig M (2010) Fitness flux and ubiquity of adaptive evolution. Proc Natl Acad Sci U S A 107: 4248–53.
- Wiley D, Wilson I, Skehel J (1981) Structural identification of the antibodybinding sites of Hong Kong inuenza haemagglutinin and their involvement in antigenic variation. Nature 289: 373.
- 40. Kryazhimskiy S, Plotkin J (2008) The population genetics of dN/dS. PLoS Genetics 4: e1000304.
- Moss WN, Priore SF, Turner DH (2011) Identification of potential conserved RNA secondary structure throughout inuenza A coding regions. RNA 17: 991– 1011.
- Rouzine I, Rodrigo A, Coffin J (2001) Transition between stochastic evolution and deterministic evolution in the presence of selection: general theory and application to virology. Microbiology and molecular biology reviews 65: 151.
- Plotkin JB, Dushoff J, Levin SA (2002) Hemagglutinin sequence clusters and the antigenic evolution of inuenza A virus. Proceedings of the National Academy of Sciences of the United States of America 99: 6263–8.
- Grenfell BT, Pybus OG, Gog JR, Wood JLN, Daly JM, et al. (2004) Unifying the epidemiological and evolutionary dynamics of pathogens. Science 303: 327– 32.
- Koelle K, Cobey S, Grenfell B, Pascual M (2006) Epochal evolution shapes the phylodynamics of interpandemic inuenza A (H3N2) in humans. Science 314: 1898–903.
- Bedford T, Rambaut A, Pascual M (2012) Canalization of the evolutionary trajectory of the human inuenza virus. BMC Biol 10: 38.
- Kryazhimskiy S, Dushoff J, Bazykin GA, Plotkin JB (2011) Prevalence of epistasis in the evolution of inuenza A surface proteins. PLoS Genetics 7: e1001301.
- Edgar RC (2004) Muscle: multiple sequence alignment with high accuracy and high throughput. Nucleic Acids Research 32: 1792–1797.